Author’s response to reviews

Title: Cross-sectional survey of knowledge of obstetric danger signs among women in rural Madagascar.

Authors:

Ania Salem (ania.salem@etu.unige.ch)
Oriane Lacour (oriane.lacour@etu.unige.ch)
Stefano Scaringella (scaringella59@gmail.com)
Josea Herinianasolo (jheriniansolo@gmail.com)
Anne-Caroline Benski (Anne-Caroline.Benski@hcuge.ch)
Giovanna Stancanelli (giovanna.stancanelli@gmail.com)
Pierre Vassilakos (pierrevassilakos@bluewin.ch)
Patrick Petignat (patrick.petignat@hcuge.ch)
Nicole Schmidt (nicole.ch.schmidt@gmail.com)

Version: 4 Date: 08 Jan 2018

Author’s response to reviews:

Dear Editor in Charge

Thank you for the recent contact and advice. Please find attached our final version of the clean manuscript.

1. Figure 1, was removed. Associated citations and figure numbering were adapted.

2. Figure legends embedded within the figure files were removed and only listed in the "Figure legends" at the end of the main manuscript.
We hope that BMC Pregnancy & Childbirth will be able to accept our manuscript for publication.

Kind regards Nicole